

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL107214

Key Points:

- Soil emission rates of NO in common inventories may be between 10× and 20× too low over the Amazon basin
- Higher soil NO over the Amazon basin alters the oxidative capacity of the atmosphere and decreases global methane lifetime by 1.4%–2.6%
- Global methane lifetime is more sensitive to Amazon soil NO fluxes than to the loss of terpene fluxes from total deforestation of the Amazon

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. L. S. Swann, aswann@uw.edu

Citation:

Lee, B. H., Munger, J. W., Wofsy, S. C., Rizzo, L. V., Yoon, J. Y. S., Turner, A. J., et al. (2024). Sensitive response of atmospheric oxidative capacity to the uncertainty in the emissions of nitric oxide (NO) from soils in Amazonia. *Geophysical Research Letters*, 51, e2023GL107214. https://doi.org/10.1029/2023GL107214

Received 22 NOV 2023 Accepted 7 MAY 2024

Author Contributions:

Conceptualization: Ben H. Lee, James Y. S. Yoon, Alexander J. Turner, Joel A. Thornton, Abigail L. S. Swann Data curation: Ben H. Lee, J. William Munger, Steven C. Wofsv Formal analysis: Ben H. Lee Funding acquisition: Abigail L. S. Swann Investigation: Ben H. Lee Methodology: Abigail L. S. Swann Project administration: Abigail L. S. Swann Resources: J. William Munger, Steven C. Wofsy, Luciana V. Rizzo Supervision: Abigail L. S. Swann Visualization: Ben H. Lee Writing - original draft: Ben H. Lee

© 2024. The Author(s). This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Sensitive Response of Atmospheric Oxidative Capacity to the Uncertainty in the Emissions of Nitric Oxide (NO) From Soils in Amazonia

Ben H. Lee¹, J. William Munger², Steven C. Wofsy², Luciana V. Rizzo³, James Y. S. Yoon¹, Alexander J. Turner¹, Joel A. Thornton¹, and Abigail L. S. Swann¹

¹University of Washington, Seattle, WA, USA, ²Harvard University, Cambridge, MA, USA, ³University of São Paulo, São Paulo, Brazil

Abstract Soils are a major source of nitrogen oxides, which in the atmosphere help govern its oxidative capacity. Thus the response of soil nitric oxide (NO) emissions to forcings such as warming or forest loss has a meaningful impact on global atmospheric chemistry. We find that the soil emission rate of NO in Amazonia from a common inventory is biased low by at least an order of magnitude in comparison to tower-based observations. Accounting for this regional bias decreases the modeled global methane lifetime by 1.4%–2.6%. In comparison, a fully deforested Amazonia, representing a 37% decrease in global emissions of isoprene, decreases methane lifetime by at most 4.6%, highlighting the sensitive response of oxidation rates to changes in emissions of NO compared to those of terpenes. Our results demonstrate that improving our understanding of soil NO emissions will yield a more accurate representation of atmospheric oxidative capacity.

Plain Language Summary Soils emit a gas called nitric oxide (NO). The amount of NO emitted from soils in tropical forests is not well known, but has been assumed to be small. We simulated how different amounts of NO emissions from soils in the Amazon impacted atmospheric chemistry. By comparing our modeled results to observations we found that NO emissions from soils in the Amazon may be between 10 and 20 times larger than the current default assumption. This matters because NO has a strong influence on the rate at which methane and other gasses are removed from the atmosphere. Therefore soil NO emissions end up impacting how long methane can last in the atmosphere, and that would have additional climate impacts. Even though soil NO emissions are smaller than other sources of NO emissions globally, we find that the amount of NO emitted in the Amazon matters a lot for how long methane can last in the atmosphere. Our results demonstrate that improving estimates of soil NO emissions is going to be necessary for making accurate estimates of how long methane and other reactive species will stay in the atmosphere.

1. Introduction

Nitrogen oxide radicals ($NO_x = NO + NO_2$) play a critical role in regulating the atmosphere's oxidizing capacity through reactions that recycle hydrogen oxides ($HO_x = OH + HO_2$) during the oxidation of hydrocarbons. Accurate representation of this chemistry is crucial, particularly in remote forested regions characterized by relatively low NO_x burdens and high emission rates of biogenic terpenes that give rise to steeply non-linear effects on HO_x , which govern the lifetimes of virtually all hydrocarbons including isoprene and methane. Notably and most recently, Wells et al. (2020) attributed the high-bias in modeled isoprene levels over Amazonia compared to those observed by the Cross-track Infrared Sounder (CrIS) to a low bias in NO, likely from soils, causing a low bias in OH, thus, a runaway increase in modeled isoprene lifetime.

The atmosphere in and above forests has long been thought to be depressed of hydroxyl (OH) radicals (Jacob & Wofsy, 1988; Logan et al., 1981; Spivakovsky et al., 2000) due to the high concentrations of volatile organic compounds (VOC), which react with OH to form organic peroxy radicals (RO₂), and limited availability of NO_x. In the absence of NO_x, RO₂ will self-react and react with hydroperoxy radicals (HO₂) to net remove OH and thus depress oxidative capacity. The availability of NO enables the catalytic recycling of OH during the daytime via reactions with RO₂ and HO₂.

 $RO_2 + NO \rightarrow RO + NO_2$



Writing – review & editing: J. William Munger, Steven C. Wofsy, Luciana V. Rizzo, James Y. S. Yoon, Alexander J. Turner, Joel A. Thornton, Abigail L. S. Swann

$$RO + O_2 \rightarrow HO_2 + R'O$$

 $HO_2 + NO \rightarrow OH + NO_2$
 $NO_2 + hv (+O_2) \rightarrow NO + O_2$

While there have been additional insights into OH recycling pathways from RO_2 chemistry which do not require NO_x , these recycling pathways are not sufficient to explain the Amazon regional observations of isoprene and formaldehyde from satellites (Wells et al., 2020). Tropical forests, and the associated large emissions of reactive VOC, are therefore significant influences on hemispheric if not global oxidizing capacity with consequences for the lifetimes of other trace gases such as methane. In-situ observations of OH and HO₂ in forested environments, however, often show that HO_x concentrations are in fact elevated above expectations (Lelieveld et al., 2008; Tan et al., 2001; Whalley et al., 2011). Implementation of updated chemical mechanisms focused on isoprene—globally the most abundantly emitted volatile organic compound (VOC) (Guenther et al., 2006; Messina et al., 2016)—have elevated modeled HO_x levels (Bates & Jacob, 2019; Squire et al., 2015; Taraborrelli et al., 2012). Recently, Jeong et al. (2022) report that measured OH during the GoAmazon campaign in the 2014 wet season near Manaus, Brazil agreed well with those calculated using multiple chemical models, suggesting that the VOC-RO₂-HO_x-NO_x coupled chemistry is well understood. However, NO measurements during that experiment were unavailable, which is emblematic of the inherent challenges of deployments to such regions where testing of our understanding of emissions and chemistry is most lacking.

The sensitivity of the burden and lifetime of methane to global OH levels is well studied (Squire et al., 2015; Voulgarakis et al., 2013; Wild et al., 2020; Zhao et al., 2020). The role that anthropogenic NO_x plays in governing OH and therefore methane levels is also well documented (Laughner et al., 2021; Peng et al., 2022; Stevenson et al., 2022). Less emphasis has been placed on the role of NO_x from natural sources, though Song et al. (2021) report that approximately half of atmospheric NO_x is now derived from sources other than fossil fuel combustion. Soils produce NO through nitrification/denitrification with subsequent emission to the atmosphere moderated by a suite of environmental conditions including soil type, gas diffusivity, moisture, temperature, etc. (Pilegaard, 2013). Alterations to forests, whether through conversion to pasture or logging, will consequently result in changes to soil emissions (Garcia-Montiel et al., 2001; Keller et al., 2005).

Numerous global chemical models—including the Community Earth System Model (CESM2) utilized here refer to the work of Yienger and Levy (1995) to constrain soil NO emissions. More recent work by Hudman et al. (2012), on which Wells et al. (2020) rely, developed an updated parameterization including a more physical representation of soil processes derived from field observations that yielded greater soil NO emissions globally (10.7 Tg N yr⁻¹) compared to that of Yienger and Levy (7.4 Tg N yr⁻¹), though rates over Amazonia are significantly lower due likely to the inclusion of NO_x loss on canopy surfaces. Previously published inventories (Davidson, 1993; Ganzeveld et al., 2002; Jaeglé et al., 2005; Martin et al., 2003; Vinken et al., 2014; Yan et al., 2005) also show disagreement over the Amazon basin additionally highlighting the uncertainty in soil emission rates of NO, the tropospheric abundance of which is not reliably inferred from space.

We present the sensitivity of the oxidative capacity of the atmosphere, and specifically, the global lifetime of methane as a function of increasing soil NO emission rates over just the Amazon region as supported by in-situ observations at the Tapajos National Forest (54.580°W, 2.51°S, Pará, Brazil) from January to August of 2015. We compare this scenario to a set of simulations in which the region is fully deforested leading to a near ceasing of the emissions of biogenic terpenes, which are known to have implications on chemistry and climate (L. Ganzeveld et al., 2010; L. Ganzeveld & Lelieveld, 2004; Geddes et al., 2016; Heald & Spracklen, 2015; Keller et al., 2005; Lathière et al., 2006; Opacka et al., 2021; Unger, 2014; Wiedinmyer et al., 2006; Wu et al., 2012). Results presented here demonstrate the importance of capturing not only the magnitude of NO emissions but responses to evolving conditions given the sensitive response of the oxidative capacity, particularly in and above forests in tropical regions where methane oxidation is likely fastest and least certain (Turner et al., 2019).

2. Methods

We performed six sets of coupled biosphere-atmospheric chemistry simulations using the CESM2 model with CAM6-chem interactive chemistry (Danabasoglu et al., 2020; Emmons et al., 2020) and with active

biogeochemistry and fires ("FCfireHIST" compset in CESM2). The model is computed at a spatial resolution of $\sim 1^{\circ}$ and a time resolution of 30 min. We use the standard set of historical emissions provided with CESM2 which were derived for the Coupled Model Intercomparison Project version 6 (CMIP6; see full details in Emmons et al., 2020). We branched our forested baseline simulation in year 1970 from the standard transient historical simulation completed with CESM2 for CMIP6 using specified sea surface temperatures. We ran our forested baseline simulation from years 1970 to 2005 and all further model runs were branched from the forested baseline in the year 2001.

We completed three separate simulations with historical forest cover in Amazonia with a forested state from 1970. We completed an additional three simulations with a deforested Amazonia that was effectively devoid of trees by changing the plant functional type to a grassland over the region (from 16°S to 8°N and from 48°W to 78°W), as illustrated in Figure S1 in Supporting Information S1. Leaf area was calculated prognostically, as were other carbon fluxes and pools, however, atmospheric CO₂ concentrations were specified based on observed concentrations for each year. The removal of trees in the model changes physical fluxes of energy and water, as well as carbon and chemical species (e.g., isoprene). The model calculated total leaf area index for the Amazon basin is about 6.5 m² m⁻² in the forested scenarios and about 1.8 m² m⁻² in the deforested scenario, wherein the emission rates of isoprene and other larger biogenic terpenes are negligible. For each of the forested and deforested Amazonia scenarios we calculated three simulations representing different soil NO conditions: (a) baseline soil NO emission rate based on the work of Yienger and Levy (1995) as included in the POET inventory (Granier et al., 2005), and factors of (b) 10× that rate, and (c) 20× of that rate. The specified NO fluxes from Yienger and Levy (1995) are disconnected from the plant types specified in CESM, and thus the specified soil NO fluxes are identical between forested and deforested simulations for the same NO perturbation.

All simulation scenarios span from 2001 to 2005, except for the forested baseline soil NO scenario which spans from 1970 through 2014. All aspects of the model except for plant type and soil NO fluxes within the Amazon vary transiently across years based on observed quantities, including greenhouse gas concentrations, sea surface temperatures, and land cover change outside the Amazon basin as specified in CMIP6 historical simulations (Eyring et al., 2016).

We also utilized F0AM (Framework for 0-D Atmospheric Modeling; Wolfe et al., 2016), which leverages the more comprehensive Master Chemical Mechanism (MCM) version 3.3.1 (Jenkin et al., 1997, 2015; Saunders et al., 2003) to compare against the CESM2 results. We performed a series of simulations. Each was conducted with constant introduction rates of NO, O₃, and isoprene into the well-mixed volume with active photochemistry ($J_{NO2} = 2 \cdot 10^{-2} \text{ s}^{-1}$), and allowed to evolve for 60 hr, much longer than the time needed for all species to reach steady state. The rate of NO introduction was varied widely to capture the response of oxidation chemistry to the steady state NO mixing ratios, while the rates of introduction of O₃ and isoprene were set such that the steady state resulting mixing ratios were comparable to the levels reported by CESM2 for the three soil NO emission rate scenarios. We compare in the section below the levels of OH and HO₂ calculated by CESM2 and F0AM.

Mixing ratios of NO, nitrogen dioxide (NO₂), and O₃ were measured at the Tapajos National Forest from January to August of 2015. The Tapajos National Forest site is located on a flat plateau and is considered a moist Tropical forest, falling in the 25th percentile for annual rainfall and 30th percentile for length of wet season (Saleska et al., 2003). Soils are primarily nutrient-poor clay oxisols with some sandy utisols (Rice et al., 2004). The site has not been recently disturbed and has minimal impact of anthropogenic air pollution (Rice et al., 2004). Ambient air in excess of instrument requirements was drawn in at 4-6 L per minute (lpm) through inlets located at eight heights off of the tall tower (0.91, 3.05, 10.42, 19.57, 28.71, 39.41, 53.04, 62.24 m above the ground). The NO chemiluminescence analyzer drew ~ 2 lpm and the O₃ analyzer drew ~ 1 lpm. To minimize sample residence time in the inlets and provide constant pressure to the analyzers, excess flow was pulled by a bypass pump to maintain constant pressure measured by a pressure controller. The inlets were sampled in sequence for 4 min each and the first 150 s after each level switch were discarded. NO and NO2 were measured by an EcoPhysics CLD-780TR analyzer equipped with an external NO₂ photolysis cell using a Hamamatsu LED with peak wavelength at 365 nm (Pollack et al., 2010). The photolysis cell was toggled on and off at 60 s intervals to provide both NO and NO + NO₂ measurement at each sample height. All instruments were housed in an air conditioned shed near the base of the tower. The NO analyzer background signal was measured at the beginning of each profile sequence by adding O₃ generated by a Hg-vapor lamp to the sample stream to convert NO to NO₂ before the sample entered the detector. Instrument gain and NO₂ conversion efficiency were determined by routinely adding a small flow of NO



Geophysical Research Letters



Figure 1. Mixing ratios of (a) NO, (b) NO₂, and (c) O_3 at the Tapajos National Forest as observed at eight heights on the tall tower from January to August of 2015, and as modeled for the lowest level of the atmosphere in CESM2 under three soil NO emission rate scenarios with baseline emissions in blue, 10× baseline in yellow and 20× baseline in green. The observational lower limit of detection for NO is plotted as the black dashed vertical line in panel (a). Panel (d) shows soil NO emission rates measured over tropical soils in unperturbed forests (Bakwin et al., 1990; Erickson et al., 2002; Garcia-Montiel et al., 2001; Gut et al., 2002; Keller et al., 2005; Verchot et al., 1999; Weitz et al., 1998). The measurements span different time periods and durations, seasons, soil type, etc. The shaded bars represent the modeled soil NO emission rate in CESM for the minimum and maximum rates over the course of a model year, with blue shading for the baseline soil NO emissions, yellow shading for 10× and green shading for 20×.

or NO_2 standard to the sample inlet. A minimum detection limit computed from twice the standard deviation of the signal during each zero check and the calibration factor measured nearest in time averaged 45 ppt over the course of the campaign. The average calibration factor uncertainty computed from standard deviations for the calibration and background signals was 10%, with most individual observations being closer to 5%. Observations of NO, NO₂, and O3 were measured every 4 min, and we averaged these high frequency values to monthly means before comparing with modeled values.

In addition to comparison with measured NO fluxes reported here we compare our CESM2 modeled quantities from observations reported previously in the literature. Observations of isoprene flux and mixing ratios were conducted at the Tapajos National Forest in June of 2016 and inferred from Sarkar et al. (2020). Observations of OH concentration were measured outside of Manaus, Brazil during GoAmazon in February–March of 2014 and inferred from Jeong et al. (2022).

3. Results and Discussion

In comparison to the measurements made above the forest canopy, CESM2 underestimates NO mixing ratio in the lower-most level of the atmosphere in the model-grid encompassing the Tapajos National Forest, when utilizing the baseline soil NO emission rate as prescribed by the work of Yienger and Levy (1995) (Figures 1a and 2a; Figure S2 in Supporting Information S1). Observed vertical profile from 0.9 to 62 m from the forest floor show that NO is most abundant near the forest floor, elevated to as high as a few parts per billion (ppb). Its mixing ratio decreases with height predominantly as it reacts with O_3 to form NO₂ (Bakwin et al., 1990), indicating that the dominant NO source at this site is likely from the soil underneath the forest canopy, and negligibly from an upwind source. The years 2015 and 2016 were characterized by El Niño conditions. The resulting drier-thannormal conditions in Amazonia could have contributed to higher than normal soil NO emissions, which exhibit a non-linear response to changing soil moisture levels (Davidson, 1993; Davidson & Kingerlee, 1997; Garcia-Montiel et al., 2001; Gut et al., 2002; Luo et al., 2012). However, El Niño became active at earliest in March of 2015, whereas the model underestimates observations throughout the year with no noticeable change in the model-observation discrepancy around that time. Biomass burning as well does not appear to have had a significant long-term impact on NO levels at the site given that the observations at the top of the tower do not exhibit periodic bursts or enhancements associated with influence from fresh or intense combustion sources (Figure S2 in Supporting Information S1).





Figure 2. Monthly means of (a) NO mixing ratio, (b) isoprene flux, (c) isoprene mixing ratio, and (d) OH concentration in the lower-most level of the atmosphere above the Tapajos National Forest. Observations of isoprene flux and mixing ratios were conducted at the Tapajos National Forest in June of 2016 and inferred from Sarkar et al. (2020), while OH concentration was measured outside of Manaus, Brazil during GoAmazon in Feb-Mar of 2014 and inferred from Jeong et al. (2022). Observed isoprene and OH values represent respective campaign mean values.

Observed NO₂ and O₃ levels are, likewise, greater than those modeled above the canopy (the lowest level of the atmosphere) when implementing the baseline soil NO emission rate. One implication is that the modeled oxidative capacity of the atmosphere, namely OH and HO₂, is also likely biased low given NO cycles HO₂ to OH and that O_3 is the dominant primary HO_x source. As such, increasing the model soil NO emission rate enhances the levels of NO_x, HO_x, and O₃, as shown in Figure 1 (and Figure S3 in Supporting Information S1). Under the baseline soil scenario (blue range in Figure 1), the model underestimates above canopy NOx much more (by ~factor 10) than it does O3. As model NO emissions are increased, NOx comes into closer agreement, while model O3 overestimates observations by not more than factor of 2. The median level of NO measured above the forest canopy generally resides between that modeled with 10× and 20× soil emission rates. Similarly, modeled soil NO emission rates at 10× and 20× also show closer agreement with measured soil NO emission rates from unperturbed tropical forests (Figure 1d), indicating that the seemingly large increase in NO of 10× and 20× is easily within the range of observations. The mean ratios of the observed to modeled NO_x (NO + NO₂) resulting from utilizing the baseline, 10×, and 20× soil emission rates are lower at 2.7, 0.7, and 0.4, respectively, because NO₂ (and O₃) is overestimated by CESM2 with increasing soil NO fluxes for reasons detailed below. We note that the inventory from which CESM2 derives the default NO fluxes (Yienger & Levy, 1995) has substantially larger fluxes over the Amazon basin compared to a more recent inventory (Hudman et al., 2012). Thus our findings that the NO fluxes from Yienger and Levy (1995) appear too low by an order of 10× and 20× over the Amazon basin is not reconciled by the use of more recent inventories.

With the enhanced soil emission rates, OH—though never measured at the Tapajos National Forest—increases to levels that are well within an order of

magnitude of that observed during the GoAmazon experiment (Jeong et al., 2022) (Figure 2d), with corresponding effects on the lifetime and burden of isoprene. However, the model overestimates isoprene mixing ratio compared to observations made at the Tapajos National Forest made in June of 2016 (Sarkar et al., 2020) by a factor between 8 and 12 (Figure 2c). This discrepancy is not necessarily due to inaccurate isoprene fluxes (Figure 2b), which Sarkar et al. (2020) finds agrees well at their site between what is measured by the eddy covariance approach and modeled using MEGAN (Model of Emissions of Gases and Aerosols from Nature) which also constrains biogenic terpene emissions for CESM2. However, Jo et al. (2021) evaluated the isoprene emissions in CESM2 and hypothesized that the emissions are too large, although they did not compare modeled quantities with direct observations from the Amazon basin.

In addition to a low bias in soil NO emission rate, mixing in the boundary layer within CESM2 may be too slow as evidenced by the overestimation of isoprene levels at the surface while model OH levels and model isoprene fluxes from the surface are reasonable. The vertical profile of modeled NO in the boundary layer exhibits a sharp enhancement in the lowest level of the atmosphere that interacts with the surface where emissions are continuously occurring (Figure S3 in Supporting Information S1) with O₃ showing a corresponding depletion comparable to what is observed from the tower beneath the forest canopy (Figure 1). Such erroneously slow vertical mixing leads to excessive accumulation of NO in the lower-most layer of the atmosphere and deprives the rest of the boundary layer of NO (and isoprene) that would otherwise lead to enhanced O₃ and HO_x levels, and therefore a shorter isoprene lifetime. For instance, while NO levels observed at Tapajos National Forest reside between those modeled with 10x and 20x soil NO flux (Figure 1a), observed NO₂ and O₃ levels are overestimated by the enhanced soil NO fluxes in CESM2 (Figures 1b and 1c). This is consistent with an over-abundance of modeled isoprene-derived peroxy radicals (RO₂) that react with NO to form NO₂, which with slow vertical mixing within the model would lead to an overproduction of O_3 . Faster boundary layer mixing in the model would lead to dilution of NO in the surface layer and would in turn require an even stronger surface NO flux than what we have implemented here in order to match what was observed at the Tapajos National Forest. As such, the soil NO emission rate we infer here is likely a lower bound on the truth due to the slow mixing in CESM2. The rate of





Figure 3. Fractional change in near-surface level (<800 hPa or < \sim 2 km) OH due to increasing soil NO emissions from the Amazon by a factor of 20 relative to the baseline scenario in CESM2 for the month of September (wet season). Note the increase in OH due to increasing soil NO emissions is evident in all seasons (Figure 2d).

vertical mixing in CESM2 is not easily tuned. Additionally, given the lack of measured vertical profiles in the boundary layer of these compounds as well as the large variance of soil NO flux in time and space (Figure 1d), the focus of this work was not to quantify the soil NO flux at this one site, but rather to provide reasonable bounds and the subsequent impact on atmospheric oxidative capacity.

As soil NO emissions were increased in CESM2, so did the atmospheric levels of OH and HO₂. However, even with surface fluxes that appear within range of observations isoprene concentrations were still overestimated in CESM2. As such, to rule out underestimation of OH and HO₂ in CESM2, we employed F0AM to provide an additional check on the HOx budget.

There is good agreement between OH and HO_2 concentrations determined for the three forested simulations of varying soil NO emission rates and those of the F0AM simulations (Figure S4 in Supporting Information S1) indicating that all significant chemical mechanisms listed in the more comprehensive MCM v3.3.1 (Jenkin et al., 1997, 2015; Saunders et al., 2003) are represented by the chemistry module of CESM2 which is more condensed. Therefore, the high-bias in isoprene mixing ratios (Figure 2c) is likely not the result of deficient chemistry in CESM2.

We also note the variability in the observed NO values, as evidenced by the large difference in the observed mean and observed median NO mixing ratios (Figure 2). This is likely due to the dependence of nitrification and denitrification that generate NO on changes to conditions such as soil humidity and soil temperature. The modeled NO levels are much less variable—denoted by the shaded trace for a given month that represent the variability between the years 2001 and 2005—since its emissions are based on seasonally varying but annually repeating climatology. The minimum and maximum soil NO flux in the model region corresponding to the Tapajos National Forest are $3.6-10.1 \ \mu$ gNO m⁻² hr⁻¹, respectively. As a result, the response of soil NO emissions to varying environmental conditions is not readily captured currently by CESM2.

The resulting increase in OH due to increasing the soil NO emission rate of Amazonia is significant enough to be globally relevant. We find that there is about a 3-fold increase in the fractional change in OH number concentration below 800 hPa in response to increasing the baseline soil emission rate by a factor of $20 \times$ (Figure 3). The factors of $10 \times$ and $20 \times$ increases in soil NO emission rates relative to baseline rates cause decreases in the global methane lifetime of 1.4% and 2.6%, respectively (Figure 4a), driven by the imposed 2.6% and 5.5% increases in





Figure 4. Impact on global (a) methane lifetime (b) soil NO emission rates and (c) biogenic isoprene emission rates due to changes in soil NO emission rate of Amazonia and/or deforestation of Amazonia, relative to the forested Amazonia scenario with baseline NO emission rate of CESM2.

the total (natural and anthropogenic) global surface NO_x emissions (Figure 4b). For context, this is approaching the amount needed to resolve the stabilization of methane observed between the years 2000 and 2007 (Dlu-gokencky et al., 2003; Lan et al., 2022), which can be explained by an approximately 4% increase in global OH levels as supported by remote-site measurements of methyl chloroform (Rigby et al., 2017; Turner et al., 2017), and which remains yet unresolved by models (Stevenson et al., 2020; Turner et al., 2019).

We compare the global methane lifetime's sensitivity to changes in Amazonian soil NO_x emissions and terpene emissions. Nearly ceasing Amazonia's emissions of biogenic terpenes including isoprene through simulated deforestation causes a 37% decrease in the global isoprene emission rate (Figure 4c) but a 2.8% decrease in the global methane lifetime. That a much larger change in regional isoprene emissions compared to regional NO emissions is required to induce a comparable change in global OH illustrates the much higher sensitivity of global OH levels and methane lifetime to the Amazonian emission rate of NO relative to that of terpenes.

Given the interactive nature of our CESM2 simulations which include chemistry, atmospheric processes, and land surface biogeochemistry there are additional feedbacks possible between deforestation, soil NO fluxes, and atmospheric chemistry. For example, water vapor is responsive to changes in temperature (less so to surface fluxes of evapotranspiration in the tropics), and secondary aerosol formation can impact cloud cover and cloud properties. We find that the totality of these feedbacks is small compared to the impact on methane lifetime of either soil NO fluxes or deforestation, resulting in an additional -0.39% to -0.67% change in methane lifetime on top of the effects of deforestation alone (-2.82%) and soil NO fluxes alone (-1.36% to -2.53%).

Lastly, simulating deforestation with a dynamic biosphere model shows its impact on soil characteristics such as moisture and soil temperature (Figure S5 in Supporting Information S1), both of which drive nitrification and denitrification processes that generate NO (Davidson et al., 2000; Garcia-Montiel et al., 2001; Luo et al., 2012), alongside a suite of other variables such as soil and tree types, extents of nitrogen input, and history of land-use (Bakwin et al., 1990; Erickson et al., 2002; Koehler et al., 2009; Pilegaard et al., 2006). These results call for the need for a dynamic soil model for NO emissions capable of incorporating the response of soil nitrogen processing to changes in environmental conditions (i.e., soil temperature, soil moisture), to adequately represent and forecast atmospheric oxidative chemistry which are currently being developed by several groups (Val Martin et al., 2023; Zaehle & Friend, 2010; Zhao et al., 2017).

4. Conclusions

Emission rates of NO from soils in CESM2 are underestimated by at least an order of magnitude compared to ground-based observations of NO_x fluxes, partitioning and vertical profiles. Factors that likely contribute to the low-bias in existing inventories include inadequate number of field measurements encompassing multiple seasons in numerous land types that provide model constraints, as well as implementation of overly aggressive NO_x loss on forest canopy surfaces. We show that correcting this low-bias in NO_x emissions enhances regional OH levels to such an extent as to be globally significant. The soil NO emission rate over Amazonia was increased by a factor of 10× to 20× relative to the rates prescribed by Yienger and Levy (Yienger & Levy, 1995) to achieve consistency with observations at the Tapajos National Forest, resulting in a global average methane lifetime decreases of 1.4% and 2.6%, respectively. The extent of NO flux underestimation may be even greater if the model utilized here suffers from slow vertical mixing of the boundary layer. The impact on the global oxidative capacity due to such increases in soil NO emissions over Amazonia is comparable to that due to the near complete ceasing of biogenic VOC emissions from the deforestation of Amazonia. Given the sensitivity of the global atmospheric oxidative capacity to relatively small changes in our current estimation of global NO_x emissions, understanding the



magnitude and sign of the response of soil NO emissions to past and future forcings including land use and land cover change are critical for assessing the lifetimes of all reactive species of the atmosphere.

Data Availability Statement

The observations made at the Tapajos National Forest as well as the model simulation output described in this paper are available at Lee et al. (2024).

References

- Bakwin, P. S., Wofsy, S. C., Fan, S.-M., Keller, M., Trumbore, S. E., & Da Costa, J. M. (1990). Emission of nitric oxide (NO) from tropical forest soils and exchange of NO between the forest canopy and atmospheric boundary layers. *Journal of Geophysical Research*, 95(D10), 16755–16764. https://doi.org/10.1029/JD095iD10p16755
- Bates, K. H., & Jacob, D. J. (2019). A new model mechanism for atmospheric oxidation of isoprene: Global effects on oxidants, nitrogen oxides, organic products, and secondary organic aerosol. Atmospheric Chemistry and Physics, 19(14), 9613–9640. https://doi.org/10.5194/acp-19-9613-2019
- Danabasoglu, G., Lamarque, J.-F., Bacmeister, J., Bailey, D. A., DuVivier, A. K., Edwards, J., et al. (2020). The Community Earth System Model Version 2 (CESM2). Journal of Advances in Modeling Earth Systems, 12(2), e2019MS001916. https://doi.org/10.1029/2019MS001916
- Davidson, E. A. (1993). Soil water content and the ratio of nitrous oxide to nitric oxide emitted from soil. In R. S. Oremland (Ed.), Biogeochemistry of global change: Radiatively active trace gases selected papers from the tenth international symposium on environmental biogeochemistry, San Francisco, August 19–24, 1991 (pp. 369–386). Springer US. https://doi.org/10.1007/978-1-4615-2812-8_20
- Davidson, E. A., Keller, M., Erickson, H. E., Verchot, L. V., & Veldkamp, E. (2000). Testing a Conceptual Model of Soil Emissions of Nitrous and Nitric Oxides: Using two functions based on soil nitrogen availability and soil water content, the hole-in-the-pipe model characterizes a large fraction of the observed variation of nitric oxide and nitrous oxide emissions from soils. *BioScience*, 50(8), 667–680. https://doi.org/10. 1641/0006-3568(2000)050[0667:TACMOS]2.0.CO;2
- Davidson, E. A., & Kingerlee, W. (1997). A global inventory of nitric oxide emissions from soils. *Nutrient Cycling in Agroecosystems*, 48(1), 37–50. https://doi.org/10.1023/A:1009738715891
- Dlugokencky, E. J., Houweling, S., Bruhwiler, L., Masarie, K. A., Lang, P. M., Miller, J. B., & Tans, P. P. (2003). Atmospheric methane levels off: Temporary pause or a new steady-state? *Geophysical Research Letters*, 30(19). https://doi.org/10.1029/2003GL018126
- Emmons, L. K., Schwantes, R. H., Orlando, J. J., Tyndall, G., Kinnison, D., Lamarque, J.-F., et al. (2020). The Chemistry Mechanism in the Community Earth System Model Version 2 (CESM2). Journal of Advances in Modeling Earth Systems, 12(4). John Wiley & Sons, Ltd. https:// doi.org/10.1029/2019MS001882
- Erickson, H., Davidson, E. A., & Keller, M. (2002). Former land-use and tree species affect nitrogen oxide emissions from a tropical dry forest. Oecologia, 130(2), 297–308. https://doi.org/10.1007/s004420100801
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Ganzeveld, L., Bouwman, L., Stehfest, E., van Vuuren, D. P., Eickhout, B., & Lelieveld, J. (2010). Impact of future land use and land cover changes on atmospheric chemistry-climate interactions. *Journal of Geophysical Research*, 115(D23), D23301. https://doi.org/10.1029/ 2010JD014041
- Ganzeveld, L., & Lelieveld, J. (2004). Impact of Amazonian deforestation on atmospheric chemistry. *Geophysical Research Letters*, 31(6), L06105. https://doi.org/10.1029/2003GL019205
- Ganzeveld, L. N., Lelieveld, J., Dentener, F. J., Krol, M. C., Bouwman, A. J., & Roelofs, G.-J. (2002). Global soil-biogenic NOx emissions and the role of canopy processes. *Journal of Geophysical Research*, 107(D16), ACH9-1–ACH9-17. https://doi.org/10.1029/2001JD001289
- Garcia-Montiel, D. C., Steudler, P. A., Piccolo, M. C., Melillo, J. M., Neill, C., & Cerri, C. C. (2001). Controls on soil nitrogen oxide emissions from forest and pastures in the Brazilian Amazon. Global Biogeochemical Cycles, 15(4), 1021–1030. https://doi.org/10.1029/2000GB001349
- Geddes, J. A., Heald, C. L., Silva, S. J., & Martin, R. V. (2016). Land cover change impacts on atmospheric chemistry: Simulating projected largescale tree mortality in the United States. Atmospheric Chemistry and Physics, 16(4), 2323–2340. https://doi.org/10.5194/acp-16-2323-2016
- Granier, C., Lamarque, J. F., Mieville, A., Muller, J. F., Olivier, J., Orlando, J., et al. (2005). POET, a database of surface emissions of ozone precursors Retrieved from http://accent.aero.jussieu.fr/Documents/POET_documentation.pdf, https://eccad3.sedoo.fr/
- Guenther, A., Karl, T., Harley, P., Wiedinmyer, C., Palmer, P. I., & Geron, C. (2006). Estimates of global terrestrial isoprene emissions using MEGAN (model of emissions of gases and aerosols from nature). Atmospheric Chemistry and Physics, 6(11), 3181–3210. https://doi.org/10. 5194/acp-6-3181-2006
- Gut, A., van Dijk, S. M., Scheibe, M., Rummel, U., Welling, M., Ammann, C., et al. (2002). NO emission from an Amazonian rain forest soil: Continuous measurements of NO flux and soil concentration. *Journal of Geophysical Research*, 107(D20), LBA24-1–LBA24-10. https://doi. org/10.1029/2001JD000521
- Heald, C. L., & Spracklen, D. V. (2015). Land use change impacts on air quality and climate. *Chemical Reviews*, 115(10), 4476–4496. https://doi.org/10.1021/cr500446g
- Hudman, R. C., Moore, N. E., Mebust, A. K., Martin, R. V., Russell, A. R., Valin, L. C., & Cohen, R. C. (2012). Steps towards a mechanistic model of global soil nitric oxide emissions: Implementation and space based-constraints. *Atmospheric Chemistry and Physics*, 12(16), 7779–7795. https://doi.org/10.5194/acp-12-7779-2012
- Jacob, D. J., & Wofsy, S. C. (1988). Photochemistry of biogenic emissions over the Amazon forest. *Journal of Geophysical Research*, 93(D2), 1477–1486. https://doi.org/10.1029/JD093iD02p01477
- Jaeglé, L., Steinberger, L., Martin, R. V., & Chance, K. (2005). Global partitioning of NOx sources using satellite observations: Relative roles of fossil fuel combustion, biomass burning and soil emissions. *Faraday Discussions*, 130(0), 407–423. https://doi.org/10.1039/B502128F
- Jenkin, M. E., Saunders, S. M., & Pilling, M. J. (1997). The tropospheric degradation of volatile organic compounds: A protocol for mechanism development. Atmospheric Environment, 31(1), 81–104. https://doi.org/10.1016/S1352-2310(96)00105-7
- Jenkin, M. E., Young, J. C., & Rickard, A. R. (2015). The MCM v3.3.1 degradation scheme for isoprene. Atmospheric Chemistry and Physics, 15(20), 11433–11459. https://doi.org/10.5194/acp-15-11433-2015

Acknowledgments

ALSS and BHL acknowledge NSF Macrosystems award to the University of Washington DEB-1925837. LVR acknowledges the Sao Paulo State Funding Agency (FAPESP 10156-8/2013). JAT acknowledges an award from the NSF Atmospheric and Geo-space Sciences AGS-2023670. Trace gas observations at the Tapajos km67 site were supported by Department of Energy Grant DE-SC0008311. We especially thank the staff at the LBA office in Santarem for logistical support.

- Jeong, D., Seco, R., Emmons, L., Schwantes, R., Liu, Y., McKinney, K. A., et al. (2022). Reconciling observed and predicted tropical rainforest OH concentrations. *Journal of Geophysical Research: Atmospheres, 127*(1), e2020JD032901. https://doi.org/10.1029/2020JD032901
- Jo, D. S., Hodzic, A., Emmons, L. K., Tilmes, S., Schwantes, R. H., Mills, M. J., et al. (2021). Future changes in isoprene-epoxydiol-derived secondary organic aerosol (IEPOX SOA) under the Shared Socioeconomic Pathways: The importance of physicochemical dependency. Atmospheric Chemistry and Physics, 21(5), 3395–3425. https://doi.org/10.5194/acp-21-3395-2021
- Keller, M., Varner, R., Dias, J. D., Silva, H., Crill, P., Oliveira, R. C., & Asner, G. P. (2005). Soil–atmosphere exchange of nitrous oxide, nitric oxide, methane, and carbon dioxide in logged and undisturbed forest in the Tapajos National Forest, Brazil. *Earth Interactions*, 9(23), 1–28. https://doi.org/10.1175/EI125.1
- Koehler, B., Corre, M. D., Veldkamp, E., Wullaert, H., & Wright, S. J. (2009). Immediate and long-term nitrogen oxide emissions from tropical forest soils exposed to elevated nitrogen input. *Global Change Biology*, 15(8), 2049–2066. https://doi.org/10.1111/j.1365-2486.2008.01826.x
- Lan, X., Thoning, K. W., & Dlugokenchy, E. J. (2022). Trends in globally-averaged CH4, N2O, and SF6 determined from NOAA global monitoring laboratory measurements (version 2023-09). [Dataset]. National Oceanic and Atmospheric Administration. https://doi.org/10. 15138/P8XG-AA10
- Lathière, J., Hauglustaine, D. A., Friend, A. D., De Noblet-Ducoudré, N., Viovy, N., & Folberth, G. A. (2006). Impact of climate variability and land use changes on global biogenic volatile organic compound emissions. *Atmospheric Chemistry and Physics*, 6(8), 2129–2146. https://doi. org/10.5194/acp-6-2129-2006
- Laughner, J. L., Neu, J. L., Schimel, D., Wennberg, P. O., Barsanti, K., Bowman, K. W., et al. (2021). Societal shifts due to COVID-19 reveal large-scale complexities and feedbacks between atmospheric chemistry and climate change. *Proceedings of the National Academy of Sciences* of the United States of America, 118(46), e2109481118. https://doi.org/10.1073/pnas.2109481118
- Lee, B., Munger, B., Wofsy, S. C., Rizzo, L. V., Yoon, J. S., Turner, A. J., et al. (2024). Data from: Sensitive response of atmospheric oxidative capacity to the uncertainty in the emissions of nitric oxide (NO) from soils in Amazonia [Dataset]. Dryad. https://doi.org/10.5061/dryad. gf1vhhmw3
- Lelieveld, J., Butler, T. M., Crowley, J. N., Dillon, T. J., Fischer, H., Ganzeveld, L., et al. (2008). Atmospheric oxidation capacity sustained by a tropical forest. *Nature*, 452(7188), 737–740. https://doi.org/10.1038/nature06870
- Logan, J. A., Prather, M. J., Wofsy, S. C., & McElroy, M. B. (1981). Tropospheric chemistry: A global perspective. Journal of Geophysical Research, 86(C8), 7210–7254. https://doi.org/10.1029/JC086iC08p07210
- Luo, G. J., Brüggemann, N., Wolf, B., Gasche, R., Grote, R., & Butterbach-Bahl, K. (2012). Decadal variability of soil CO₂, NO, N₂O, and CH₄ fluxes at the Höglwald Forest, Germany. *Biogeosciences*, 9(5), 1741–1763. https://doi.org/10.5194/bg-9-1741-2012
- Martin, R. V., Jacob, D. J., Chance, K., Kurosu, T. P., Palmer, P. I., & Evans, M. J. (2003). Global inventory of nitrogen oxide emissions constrained by space-based observations of NO₂ columns. *Journal of Geophysical Research*, 108(D17). https://doi.org/10.1029/2003JD003453
- Messina, P., Lathière, J., Sindelarova, K., Vuichard, N., Granier, C., Ghattas, J., et al. (2016). Global biogenic volatile organic compound emissions in the ORCHIDEE and MEGAN models and sensitivity to key parameters. *Atmospheric Chemistry and Physics*, 16(22), 14169–14202. https://doi.org/10.5194/acp-16-14169-2016
- Opacka, B., Müller, J.-F., Stavrakou, T., Bauwens, M., Sindelarova, K., Markova, J., & Guenther, A. B. (2021). Global and regional impacts of land cover changes on isoprene emissions derived from spaceborne data and the MEGAN model. *Atmospheric Chemistry and Physics*, 21(11), 8413–8436. https://doi.org/10.5194/acp-21-8413-2021
- Peng, S., Lin, X., Thompson, R. L., Xi, Y., Liu, G., Hauglustaine, D., et al. (2022). Wetland emission and atmospheric sink changes explain methane growth in 2020. *Nature*, 612(7940), 477–482. https://doi.org/10.1038/s41586-022-05447-w
- Pilegaard, K. (2013). Processes regulating nitric oxide emissions from soils. Philosophical Transactions of the Royal Society B: Biological Sciences. Royal Society. https://doi.org/10.1098/rstb.2013.0126
- Pilegaard, K., Skiba, U., Ambus, P., Beier, C., Brüggemann, N., Butterbach-Bahl, K., et al. (2006). Factors controlling regional differences in forest soil emission of nitrogen oxides (NO and N₂O). *Biogeosciences*, 3(4), 651–661. https://doi.org/10.5194/bg-3-651-2006
- Pollack, I. B., Lerner, B. M., & Ryerson, T. B. (2010). Evaluation of ultraviolet light-emitting diodes for detection of atmospheric NO2 by photolysis - Chemiluminescence. *Journal of Atmospheric Chemistry*, 65(2), 111–125. https://doi.org/10.1007/s10874-011-9184-3
- Rice, A. H., Pyle, E. H., Saleska, S. R., Hutyra, L., Palace, M., Keller, M., et al. (2004). Carbon balance and vegetation dynamics in an old-growth Amazonian Forest. *Ecological Applications*, 14(sp4), 55–71. https://doi.org/10.1890/02-6006
- Rigby, M., Montzka, S. A., Prinn, R. G., White, J. W. C., Young, D., O'Doherty, S., et al. (2017). Role of atmospheric oxidation in recent methane growth. Proceedings of the National Academy of Sciences of the United States of America, 114(21), 5373–5377. https://doi.org/10.1073/pnas. 1616426114
- Saleska, S. R., Miller, S. D., Matross, D. M., Goulden, M. L., Wofsy, S. C., da Rocha, H. R., et al. (2003). Carbon in amazon forests: Unexpected seasonal fluxes and disturbance-induced losses. *Science*, 302(5650), 1554–1557. https://doi.org/10.1126/science.1091165
- Sarkar, C., Guenther, A. B., Park, J.-H., Seco, R., Alves, E., Batalha, S., et al. (2020). PTR-TOF-MS eddy covariance measurements of isoprene and monoterpene fluxes from an eastern Amazonian rainforest. *Atmospheric Chemistry and Physics*, 20(12), 7179–7191. https://doi.org/10. 5194/acp-20-7179-2020
- Saunders, S. M., Jenkin, M. E., Derwent, R. G., & Pilling, M. J. (2003). Protocol for the development of the master chemical mechanism, MCM v3 (Part A): Tropospheric degradation of non-aromatic volatile organic compounds. *Atmospheric Chemistry and Physics*, 3(1), 161–180. https://doi.org/10.5194/acp-3-161-2003
- Song, W., Liu, X.-Y., Hu, C.-C., Chen, G.-Y., Liu, X.-J., Walters, W. W., et al. (2021). Important contributions of non-fossil fuel nitrogen oxides emissions. *Nature Communications*, 12(1), 243. https://doi.org/10.1038/s41467-020-20356-0
- Spivakovsky, C. M., Logan, J. A., Montzka, S. A., Balkanski, Y. J., Foreman-Fowler, M., Jones, D. B. A., et al. (2000). Three-dimensional climatological distribution of tropospheric OH: Update and evaluation. *Journal of Geophysical Research*, 105(D7), 8931–8980. https://doi. org/10.1029/1999JD901006
- Squire, O. J., Archibald, A. T., Griffiths, P. T., Jenkin, M. E., Smith, D., & Pyle, J. A. (2015). Influence of isoprene chemical mechanism on modelled changes in tropospheric ozone due to climate and land use over the 21st century. *Atmospheric Chemistry and Physics*, 15(9), 5123–5143. https://doi.org/10.5194/acp-15-5123-2015
- Stevenson, D. S., Derwent, R. G., Wild, O., & Collins, W. J. (2022). COVID-19 lockdown emission reductions have the potential to explain over half of the coincident increase in global atmospheric methane. *Atmospheric Chemistry and Physics*, 22(21), 14243–14252. https://doi.org/10. 5194/acp-22-14243-2022
- Stevenson, D. S., Zhao, A., Naik, V., O'Connor, F. M., Tilmes, S., Zeng, G., et al. (2020). Trends in global tropospheric hydroxyl radical and methane lifetime since 1850 from AerChemMIP. Atmospheric Chemistry and Physics, 20(21), 12905–12920. https://doi.org/10.5194/acp-20-12905-2020

- Tan, D., Faloona, I., Simpas, J. B., Brune, W., Shepson, P. B., Couch, T. L., et al. (2001). HO x budgets in a deciduous forest: Results from the PROPHET summer 1998 campaign. *Journal of Geophysical Research*, 106(D20), 24407–24427. https://doi.org/10.1029/2001JD900016 Taraborrelli, D., Lawrence, M. G., Crowley, J. N., Dillon, T. J., Gromov, S., Groß, C. B. M., et al. (2012). Hydroxyl radical buffered by isoprene
- oxidation over tropical forests. *Nature Geoscience*, 5(3), 190–193. https://doi.org/10.1038/ngeo1405
- Turner, A. J., Frankenberg, C., & Kort, E. A. (2019). Interpreting contemporary trends in atmospheric methane. Proceedings of the National Academy of Sciences, 116(8), 2805–2813. https://doi.org/10.1073/pnas.1814297116
- Turner, A. J., Frankenberg, C., Wennberg, P. O., & Jacob, D. J. (2017). Ambiguity in the causes for decadal trends in atmospheric methane and hydroxyl. Proceedings of the National Academy of Sciences of the United States of America, 114(21), 5367–5372. https://doi.org/10.1073/pnas. 1616020114
- Unger, N. (2014). Human land-use-driven reduction of forest volatiles cools global climate. *Nature Climate Change*, 4(10), 907–910. https://doi.org/10.1038/nclimate2347
- Val Martin, M., Blanc-Betes, E., Fung, K. M., Kantzas, E. P., Kantola, I. B., Chiaravalloti, I., et al. (2023). Improving nitrogen cycling in a land surface model (CLM5) to quantify soil N₂O, NO, and NH₃ emissions from enhanced rock weathering with croplands. *Geoscientific Model Development*, 16(20), 5783–5801. https://doi.org/10.5194/gmd-16-5783-2023
- Verchot, L. V., Davidson, E. A., Cattânio, H., Ackerman, I. L., Erickson, H. E., & Keller, M. (1999). Land use change and biogeochemical controls of nitrogen oxide emissions from soils in eastern Amazonia. *Global Biogeochemical Cycles*, 13(1), 31–46. https://doi.org/10.1029/ 1998GB900019
- Vinken, G. C. M., Boersma, K. F., Maasakkers, J. D., Adon, M., & Martin, R. V. (2014). Worldwide biogenic soil NOx emissions inferred from OMI NO2 observations. Atmospheric Chemistry and Physics, 14(18), 10363–10381. https://doi.org/10.5194/acp-14-10363-2014
- Voulgarakis, A., Naik, V., Lamarque, J.-F., Shindell, D. T., Young, P. J., Prather, M. J., et al. (2013). Analysis of present day and future OH and methane lifetime in the ACCMIP simulations. *Atmospheric Chemistry and Physics*, 13(5), 2563–2587. https://doi.org/10.5194/acp-13-2563-2013
- Weitz, A. M., Veldkamp, E., Keller, M., Neff, J., & Crill, P. M. (1998). Nitrous oxide, nitric oxide, and methane fluxes from soils following clearing and burning of tropical secondary forest. *Journal of Geophysical Research*, 103(D21), 28047–28058. https://doi.org/10.1029/ 98JD02144
- Wells, K. C., Millet, D. B., Payne, V. H., Deventer, M. J., Bates, K. H., de Gouw, J. A., et al. (2020). Satellite isoprene retrievals constrain emissions and atmospheric oxidation. *Nature*, 585(7824), 225–233. https://doi.org/10.1038/s41586-020-2664-3
- Whalley, L. K., Edwards, P. M., Furneaux, K. L., Goddard, A., Ingham, T., Evans, M. J., et al. (2011). Quantifying the magnitude of a missing hydroxyl radical source in a tropical rainforest. Atmospheric Chemistry and Physics, 11(14), 7223–7233. https://doi.org/10.5194/acp-11-7223-2011
- Wiedinmyer, C., Tie, X., Guenther, A., Neilson, R., & Granier, C. (2006). Future changes in biogenic isoprene emissions: How might they affect regional and global atmospheric chemistry? *Earth Interactions*, 10(3), 1–19. https://doi.org/10.1175/EI174.1
- Wild, O., Voulgarakis, A., O'Connor, F., Lamarque, J.-F., Ryan, E. M., & Lee, L. (2020). Global sensitivity analysis of chemistry–climate model budgets of tropospheric ozone and OH: Exploring model diversity. *Atmospheric Chemistry and Physics*, 20(7), 4047–4058. https://doi.org/10. 5194/acp-20-4047-2020
- Wolfe, G. M., Marvin, M. R., Roberts, S. J., Travis, K. R., & Liao, J. (2016). The Framework for 0-D Atmospheric Modeling (F0AM) v3.1. Geoscientific Model Development, 9(9), 3309–3319. https://doi.org/10.5194/gmd-9-3309-2016
- Wu, S., Mickley, L. J., Kaplan, J. O., & Jacob, D. J. (2012). Impacts of changes in land use and land cover on atmospheric chemistry and air quality over the 21st century. Atmospheric Chemistry and Physics, 12(3), 1597–1609. https://doi.org/10.5194/acp-12-1597-2012
- Yan, X., Ohara, T., & Akimoto, H. (2005). Statistical modeling of global soil NOx emissions. Global Biogeochemical Cycles, 19(3). https://doi. org/10.1029/2004GB002276
- Yienger, J. J., & Levy, H., II. (1995). Empirical model of global soil-biogenic NOχ emissions. Journal of Geophysical Research, 100(D6), 11447–11464. https://doi.org/10.1029/95JD00370
- Zaehle, S., & Friend, A. D. (2010). Carbon and nitrogen cycle dynamics in the O-CN land surface model: 1. Model description, site-scale evaluation, and sensitivity to parameter estimates. *Global Biogeochemical Cycles*, 24(1). https://doi.org/10.1029/2009GB003521
- Zhao, Y., Saunois, M., Bousquet, P., Lin, X., Berchet, A., Hegglin, M. I., et al. (2020). On the role of trend and variability in the hydroxyl radical (OH) in the global methane budget. *Atmospheric Chemistry and Physics*, 20(21), 13011–13022. https://doi.org/10.5194/acp-20-13011-2020
- Zhao, Y., Zhang, L., Tai, A. P. K., Chen, Y., & Pan, Y. (2017). Responses of surface ozone air quality to anthropogenic nitrogen deposition in the Northern Hemisphere. Atmospheric Chemistry and Physics, 17(16), 9781–9796. https://doi.org/10.5194/acp-17-9781-2017